CS3233 Homework 5

Due on October 2 2000 at 10:00am 100 marks Topics: Backtracking and Branch and Bound

1. You are to submit your solutions to the Spanish Judge. Only answers that are accepted by the Spanish Judge are awarded marks. Pick any two questions from the following list of ten questions from the Spanish Judge. The list of questions is: f 140, 167, 179, 185, 193, 291, 307, 310, 387, 399 g. Submit your solutions to the Spanish Judge.

Each right answer is worth 50 marks. [100]

CS3233 Homework 6

Due on October 9 2000 at 10:00am 100 marks Topics: Easier Problems on the Spanish Judge

1. You are to submit your solutions to the Spanish Judge. Only answers that are accepted by the Spanish Judge are awarded marks. Pick any two questions from the following list of questions from the Spanish Judge. The list of questions is: f 136, 138, 152, 160, 227, 299, 320, 340, 357, 382, 391, 392, 408, 412, 414, 439, 441, 444, 445, 455, 458, 460, 485, 488 g. Submit your solutions to the Spanish Judge.

Each right answer is worth 50 marks. [100]